

Learning About Sea Level Rise Uncertainty Improves Coastal Adaptation Decisions

Vanessa Völz, Jochen Hinkel, Sunna Kupfer, Leigh R Macpherson, Carl Jacob

Wulff Norrby

► To cite this version:

Vanessa Völz, Jochen Hinkel, Sunna Kupfer, Leigh R
 Macpherson, Carl Jacob Wulff Norrby. Learning About Sea Level Rise Uncertainty Improves Coastal Adaptation Decisions. Earth's Future, 2024, 12 (10), 10.1029/2024ef004704 .
 hal-04928372

HAL Id: hal-04928372 https://hal.science/hal-04928372v1

Submitted on 4 Feb 2025

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License





RESEARCH ARTICLE

10.1029/2024EF004704

Special Collection:

Regional Sea Level Change and Society

Key Points:

- We develop adaptive adaptation pathways that incorporate learning about sea level rise uncertainty based on future observations
- Adaptive adaptation pathways generate sea level rise thresholds for adaptation actions that can be similar across climate change scenarios
- Adaptive adaptation pathways can reduce expected costs compared to non-adaptive pathways by up to 1.8% in our study

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to: V. Völz, vanessa.voelz@globalclimateforum.org

Citation:

Völz, V., Hinkel, J., Kupfer, S., MacPherson, L. R., & Wulff Norrby, C. J. (2024). Learning about sea level rise uncertainty improves coastal adaptation decisions. *Earth's Future*, *12*, e2024EF004704. https://doi.org/10.1029/ 2024EF004704

Received 7 APR 2024 Accepted 6 SEP 2024

© 2024. The Author(s).

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Learning About Sea Level Rise Uncertainty Improves Coastal Adaptation Decisions

Vanessa Völz^{1,2}, Jochen Hinkel^{1,2}, Sunna Kupfer³, Leigh R. MacPherson⁴, and Carl Jacob Wulff Norrby³

¹Humboldt-Universität zu Berlin, Thaer-Institute of Agricultural and Horticultural Sciences, Berlin, Germany, ²Global Climate Forum, Berlin, Germany, ³Department of Geography, Coastal Risks and Sea Level Rise Research Group, Christian-Albrechts University, Kiel, Germany, ⁴Faculty of Agricultural and Environmental Sciences, University of Rostock, Rostock, Germany

Abstract Adaptive decision-making allows decision-makers to plan long-term coastal infrastructure under uncertain sea level rise projections. To date, economic assessments of adaptive decision-making that take into account future learning about sea level rise uncertainty are rare and the existing ones have relied on simple quantification of future learning not validated against sea level science. To address this gap, we develop an economic adaptive decision-making framework that takes into account future learning about sea level rise uncertainty and apply it to a coastal case study in Lübeck, Germany, to answer the question of how adaptation to sea level rise can be improved through adaptive adaptation pathways as opposed to non-adaptive pathways. To address this question, we use a Markov decision process to formulate the stochastic optimization problem. We quantify future learning about sea level rise uncertainty through sea level rise learning scenarios based on and validated against the latest scenarios of the Intergovernmental Panel on Climate Change. Our case study results show that the city of Lübeck is currently under-protected against storm surges and that immediate adaptation actions are advisable in the face of future sea level rise. We find that adaptive adaptation pathways, in contrast to non-adaptive pathways, generate sea level rise thresholds for adaptation actions that are similar across climate change scenarios and can reduce expected costs up to 1.8%.

Plain Language Summary Climate change is causing sea levels to rise as land ice melts under higher air temperatures. By 2100, sea levels are projected to rise between 28 and 101 cm, depending on the extent of future global warming. This will increase the risk of coastal flooding in the future and require adaptation actions. Planning for coastal adaptation to sea level rise is challenged by long-lived protective infrastructure and uncertain projections of sea level rise. Adaptive decision-making methods that specifically incorporate future learning about the uncertainty of sea level rise can address this challenge. For example, observing 30 cm of sea level rise in 2060 will lead to different projections from 2050 onwards and require different adaptation actions than observing 70 cm of sea level rise in 2060. To date, economic adaptation studies that account for future learning about sea level rise uncertainty are rare, and those that do exist have relied on simple methods. We develop a decision framework to address this research gap and apply it to the city of Lübeck on the Baltic Sea in Germany. Our results show that the city of Lübeck is currently under-protected against storm surges and that immediate adaptation actions are advisable from an economic perspective.

1. Introduction

Large uncertainties in sea level rise projections, together with the long lead times and lifetimes of adaptation measures, pose a challenge to coastal adaptation planning (Fox-Kemper et al., 2021) that can be addressed through *adaptive decision-making* (Oppenheimer et al., 2019). Adaptive decision-making is an approach to decision making, where decisions are taken at multiple stages over time and decisions at future stages take into account what will be learned until then (Bellman, 1966). A number of different normative frameworks for adaptive decision-making have been proposed in recent years, including adaptive planning (Walker et al., 2001), dynamic adaptive policy pathways (Haasnoot et al., 2013) and adaptation pathway analysis (Haasnoot et al., 2012, 2020). Normative adaptive decision-making also includes some technically sophisticated methods, such as real-option analysis (Wreford et al., 2020) and dynamic planning frameworks (Herman et al., 2020), which can identify strategies that are economically efficient in terms of the timing of adaptation, taking into account future learning about the uncertainty of sea level rise (Oppenheimer et al., 2019).



As adaptive decision-making considers future learning, the results of *adaptive decision analysis methods* are *adaptive adaptation pathways*, rather than a single optimal action to be taken today, or a sequence of actions over time. Adaptive adaptation pathways, often referred to simply as adaptation pathways (Haasnoot et al., 2012), tell the decision-maker which adaptation measures to implement under which future observations of sea level rise. The majority of economic studies on coastal adaptation use a *single-stage non-adaptive adaptation pathway* (e.g., cost-benefit analysis (Anthoff et al., 2010; Bachner et al., 2022; Hallegatte et al., 2013; Hinkel et al., 2014; Lincke & Hinkel, 2018, 2021; Sassone & Schaffer, 1978; Tiggeloven et al., 2020; van Dantzig, 1956; Vousdoukas et al., 2020), which are characterized as decisions that are implemented today and remain unchanged over the entire time horizon (e.g., a constant dike height). Very few cost-benefit analyses of coastal adaptation use optimal control theory to consider a *multi-stage non-adaptive adaptation pathway* (Eigenraam et al., 2014, 2017; Zwaneveld et al., 2018), which may change over time, but the whole sequence of protection levels over time is decided today, regardless of future observations of sea level rise (e.g., a dike with incremental height upgrades).

Economically efficient adaptive adaptation pathways for coastal adaptation have rarely been developed in the literature and if so rely on simple quantification of future learning about sea level rise (Oppenheimer et al., 2019; Völz & Hinkel, 2023b). One reason for this is the unavailability of so called learning scenarios, which are a prerequisite for developing efficient adaptive adaptation pathways. Learning scenarios are an up-front quantification of the opportunity to learn about the uncertainty of sea level rise in the future (Hinkel et al., 2019; Völz & Hinkel, 2023a). While sophisticated learning scenarios and case study applications exist for precipitation (Dittrich et al., 2019; Fletcher, Lickley, & Strzepek, 2019) and groundwater level (Fletcher, Strzepek, et al., 2019), applications for sea level rise (Diaz, 2016; Garner & Keller, 2018; Gersonius et al., 2013; Linquiti & Vonortas, 2012; van der Pol et al., 2013; Webster et al., 2008; Woodward et al., 2014) use learning scenarios that are rather simple, for example, based on ad-hoc assumptions and not validated against the science of sea level rise (Völz & Hinkel, 2023b). Recently, Völz and Hinkel (2023b) have developed sea level rise learning scenarios based on and validated against sea level rise projections from the sixth assessment report (AR6) of the Intergovernmental Panel on Climate Change (IPCC), but a case study application remains to be seen.

The novelty of our paper is threefold: (a) for the first time we apply adaptive decision-making to a real case study using state-of-the-art sea level rise learning scenarios, (b) for the first time we couple a hydrodynamic flood model with an adaptive decision framework, and (c) for the first time we derive adaptive adaptation pathways with granular sea level rise thresholds for adaptation actions by optimizing timing and protection height. Recent studies on adaptive decision-making have presented optimal adaptive adaptation pathways with climate variable thresholds for binary adaptation actions that are flexible in time, as traditionally done in real-option analysis or engineering options analysis (Fletcher, Lickley, & Strzepek, 2019; Fletcher, Strzepek, et al., 2019). In the context of coastal adaptation decisions, adaptation options are usually not binary, but require a specific protection height of the coastal protection measure. While a few studies on economic adaptive decision-making combine simplified learning about climate change with optimization of timing and protection heights (Linquiti & Vonortas, 2012; van der Pol et al., 2013; Woodward et al., 2014), no study has presented optimal adaptive adaptation pathways with granular climate variable thresholds for adaptation actions with specific protection heights. We fill this gap by using a Markov decision process and state-of-the-art sea level rise learning scenarios to formulate a stochastic optimization problem and solve it using dynamic programming.

By applying our decision framework to a coastal case study in Germany, we aim to answer the following research question: How can adaptation to sea level rise be improved through economic adaptive decision analysis methods that incorporate future learning about sea level rise uncertainty, as opposed to non-adaptive decision analysis methods?

2. Methods

Our workflow for deriving optimal adaptive adaptation pathways for coastal adaptation to sea level rise consists of three steps: (a) modeling flood damages under a wide range of sea level rise, storm surges and adaptation options, (b) using the model output to build a statistical emulator of flood damages as the numerical model is computationally too expensive to be used in the decision framework, and (c) developing and applying an appropriate economic adaptive decision framework (Figure 1). In this section, we first describe the adaptive decision framework and then present its application to the German case study, including the flood damage modeling and emulation.

10.1029/2024EF004704





Figure 1. Workflow chart of our methods described in chapter 2.

2.1. Formulation of the Decision Framework

We use a Markov decision process to formulate our decision framework for coastal adaptation decision-making and apply dynamic programming to find optimal solutions. We create a version of the decision framework that incorporates learning about sea level rise uncertainty (Section 2.1.5) and, for comparison, another version that does not (Section 2.1.4).

2.1.1. Markov Decision Process

To model adaptation actions against increasing flood risks due to rising sea levels, we define a time-dependent state space S_t as a set of tuples containing sea level rise states l_t and adaptation states d_t , both from discrete sets \mathcal{L}_t and \mathcal{D}_t :

$$s_t = (l_t, d_t) \in \mathcal{L}_t \times \mathcal{D}_t = \mathcal{S}_t.$$
(1)

The state space *S*, spanning the whole time horizon of discrete time steps $\mathcal{T} = \{1, ..., T\}$, is defined by $S = \bigcup_{t \in \mathcal{T}} S_t$. In each time step, an action from a discrete set of actions \mathcal{U} can be applied. The transition function $P(s_{t+1} = s' | s_t = s, u_t = u)$ defines the likelihood of transitioning from state *s* at time *t* to *s'* when action *u* is implemented in *t*. The transition function for state s_t is a combination of two transition functions for the sea level rise and adaptation states. The transition function for sea level rise states is stochastic and independent of action *u* and will be defined in Section 2.1.5, whereas the transition function for states by interpreting *d* as the protection height and *u* as a height upgrade:

$$P(s_{t+1} = s' | s_t = s, u_t = u')$$

$$= P((l_{t+1}, d_{t+1}) = (l^k, d^q) | (l_t, d_t) = (l^j, d^r), u_t = u')$$

$$= \begin{cases} P(l_{t+1} = l^k | l_t = l^j) & \text{if } d^q = d^r + u' \\ 0 & \text{else} \end{cases}$$
(2)

Let C(s, u) be a cost function that describes all costs occurring in state *s* under action *u*. The following collection of objects constitutes a Markov decision process that describes an adaptive adaptation problem:

23284277, 2024, 10, Downloaded from https://agupub

com/doi/10.1029/2024EF004704 by Cochrane France

, Wiley Online Library

on [04/02/2025]

See

the Term

and Co

iditions (https://online.ibrary

Wiley Online Library for rules

$$\{\mathcal{T}, S, \mathcal{U}, P(s_{t+1} = s' | s_t = s, u = u'), C(s, u)\}.$$
 (3)

We want to find a adaptive adaptation pathway that tells us which adaptation action $u \in \mathcal{U}$ should be implemented for each state s_t over time. Hence, a adaptive adaptation pathway (also called decision rule or policy function) π is a function mapping from the state space to the action space: $\pi : S \mapsto \mathcal{U}$. The set of all possible adaptive adaptation pathways is defined by:

$$\Pi = \{ \pi : s \mapsto u \mid s \in \mathcal{S}, \ u \in \mathcal{U} \}.$$

$$\tag{4}$$

2.1.2. Objective Function

We aim to find an optimal adaptive adaptation pathway that minimizes costs occurring today and in future moments in time. We discount all future costs with the discount rate β to represent costs in present values, or in other words to consider time preferences for cash flows. Our framework contains stochastic sea level rise development (see Section 2.1.5) and therefore we minimize expected costs in our objective function:

$$\min_{\pi \in \Pi} \mathbb{E} \sum_{t=1}^{T} \beta^{t} C(s_{t}, \pi_{t}(s_{t})).$$
(5)

We want to consider costs arising from flood damages and adaptation investments in the cost function C(s, u). To specify the cost function in more detail, let I(d, u) be the investment cost function that defines the costs for implementing adaptation action u under current adaptation d. To account for flood damages, we use a function $\phi(l + x, d)$ that quantifies the total flood damage under sea level rise l, extreme water level x and adaptation d. We determine the average annual flood damage under sea level rise l and adaptation d by considering all possible extreme water levels and their annual likelihood of occurrence. Specifically, we multiply the flood damages by the probability density function f(x) of extreme water level x and integrate over all extreme water levels:

$$\min_{\pi \in \Pi} \mathbb{E} \sum_{t=1}^{T} \beta^{t} \bigg[I(d_{t}, \pi_{t}(s_{t})) + \int_{0}^{x_{\max}} \phi(l_{t} + x, d_{t}) f(x) \, dx \bigg].$$
(6)

2.1.3. Solving the Optimization Problem: Bellman Equation

The objective function formulated in Equation 6 is a sequential decision problem with an optimal substructure, which allows us to solve this problem step by step rather than all at once. If we are in a given state in a given time and know the future values of being in each possible future state, we would simply need to choose the optimal action at that time, taking into account the one-period contribution and the values of the future states (Powell, 2011). Thus, we can replace the minimization over all adaptive adaptation pathways in Equation 6 with the minimization over all actions for each subproblem. We assign values V_t to each state at each time step and obtain the optimality equation, often called *Bellman equation* (Puterman, 1994):

$$V_t(s_t) = \min_{u \in \mathcal{U}} \left\{ I(d_t, u) + \int_0^{x_{\max}} \phi(l_t + x, d_t) f(x) \, dx + \beta \, \mathbb{E}[V_{t+1}(s_{t+1})] \right\}$$
(7)

for $s_t \in S_t$ and t = 1, ..., T - 1. We abbreviate $P(s_{t+1} = s' | s_t = s, u_t = u)$ with P(s' | s, u) and write:

$$V_{t}(s_{t}) = \min_{u \in U} \left\{ I(d_{t}, u) + \int_{0}^{x_{\max}} \phi(l_{t} + x, d_{t}) f(x) \, dx + \beta \sum_{s' \in S_{t+1}} P(s'|s_{t}, u) \, V_{t+1}(s') \right\}.$$
(8)

The Bellman Equation 8 is the basis for finding optimal adaptive adaptation pathways for the objective function Equation 6. For stochastic finite-horizon discrete-time Markov decision problems with time-independent action space, the Bellman equation can be solved by dynamic programming (Powell, 2011). The dynamic programming algorithms used in this study are described in Supporting Information S1.



Earth's Future



Figure 2. Illustrations of the adaptation state space, sea level rise state space and resulting adaptation pathway for single-stage non-adaptive, multi-stage non-adaptive and adaptive decision analysis methods. The non-adaptive sea level rise state space (b and e) is based on probability distribution functions per time step. The adaptive sea level rise state space (h) is a learning scenario.

2.1.4. Non-Adaptive Decision Analysis Methods

To illustrate the benefits of adaptive decision analysis methods that incorporate learning about sea level rise uncertainty, we compare their results with those of non-adaptive decision analysis methods, which do not incorporate learning about sea level rise uncertainty. We define *single-stage non-adaptive decision analysis methods* as those that analyse single-stage decisions (sometimes called one-shot decisions), that is, decisions that are implemented today and remain unchanged over the entire time horizon (Figures 2a–2c). In the literature, this method is the most widely used method for the economic analysis of coastal protection measures (Anthoff et al., 2010; Bachner et al., 2022; Hallegatte et al., 2013; Hinkel et al., 2014; Lincke & Hinkel, 2018, 2021; Tiggeloven et al., 2020; van Dantzig, 1956; Vousdoukas et al., 2020). We define *multi-stage non-adaptive decision analysis methods* as those that analyse multi-stage decisions, that is, decisions where protection measures may change over time, but decisions are non-adaptive, that is, the whole sequence of protection levels over time is decided today (Figures 2d–2f). In the literature, examples of this can be found in the Dutch context in the form of cost-benefit analyses using optimal control theory (Eijgenraam et al., 2014, 2017; Zwaneveld et al., 2018).

For non-adaptive methods we do not use a sea level rise learning scenario, but instead apply a convolution of the sea level rise and extreme water level probability distribution functions and use the probability density function of this convolution for f(x) in Equation 6. We determine the convolution by a discrete sampling of the AR6 sea level rise p-box scenario for each time step with 10,000 data points and a sampling of the extreme water level distribution function, respectively, and sum both samples. Similar to Lickley et al. (2014), we then fit a log-normal distribution to the summed sample to determine an analytical solution to the convolution, see Supporting Information S1 for details.

2.1.5. Adaptive Decision Analysis Method

Uncertainty in sea level rise projections will evolve over time as new observations of sea level rise become available. Sea level rise observations will not only provide the exact value of sea level rise at that point in time,



they can also be used to learn about the uncertainty in sea level rise as sea level rise projections are updated. For example, observing 20 cm of sea level rise in 2050 will lead to different sea level rise projections from 2050 onwards than observing 40 cm of sea level rise in 2050. We define *adaptive decision analysis methods* as methods that involve future learning about sea level rise uncertainty through multi-stage adaptation actions and adaptivity (Figures 2g–2i).

In our decision framework, learning about sea level rise uncertainty is captured within the sea level rise state space, that is, the sea level rise state set \mathcal{L}_t and the sea level rise transition function $P(l_{t+1} = l^k | l_t = l^j)$. The combination of the two is a learning scenario (Figure 2h).

We apply the direct fit method by Völz and Hinkel (2023b) to define the sea level rise state set and the transition function for sea level rise. The direct fit method enumerates sea level rise states per time step and generates a binomial and recombining scenario lattice with a 50% probability of going up or down in each state. The direct fit method determines the sea level rise states positions by positioning each node in accordance with its occurrence probability in the lattice (seen from the first time step and based on the binomial distribution) and the quantile function of the underlying sea level rise data in each respective time step. For example, the two sea level rise states in the second time step both have an occurrence probability of 50% and are positioned at the 25th and 75th quantile value of the underlying sea level rise data in the year 2030. The transition function for sea level rise is then given by

$$P(l_{t+1} = l^k \mid l_t = l^j) = \begin{cases} 0.5 & \text{if } (k = j \lor k = j+1) \\ 0 & \text{else} \end{cases}$$
(9)

Inserting Equation 9 in Equation 2 gives the joint transition function

$$P(s_{t+1} = s' \mid s_t = s, u = u')$$

$$= P((l_{t+1}, d_{t+1}) = (l^k, d^q) \mid (l_t, d_t) = (l^j, d^r), u = u')$$

$$= \begin{cases} 0.5 & \text{if } (k = j \lor k = j + 1) \land d^q_{t+1} = d^r_t + u' \\ 0 & \text{else} \end{cases}$$
(10)

Note that while the structure of the learning scenario is identical to a binomial and recombining scenario lattice (sometimes called binary decision tree) used in traditional real-option analysis, the underlying generation method for positioning its nodes is novel and different from existing methods. The existing sea level rise learning scenarios are based on random sampling of distribution parameters (Woodward et al., 2014), a stochastic process (Gersonius et al., 2012), updating distribution parameters (Linquiti & Vonortas, 2012), simple physical models (Webster et al., 2008), linear model regression (Garner & Keller, 2018), or ad-hoc assumptions (van der Pol et al., 2013) and are not based on state-of-the-art sea level rise projections, for example, from AR5 or AR6, and neither of these studies validated their learning scenarios against underlying sea level rise data (Völz & Hinkel, 2023b). Völz and Hinkel (2023b) validate their sea level rise learning scenarios against AR6 projections and show that the scenarios are able to reasonably represent sea level rise development over time and future learning from observations. Further, the binomial and recombining scenario lattice structure of the direct fit method outperforms existing sophisticated learning scenarios in terms of computational efficiency, as non-recombining (Erfani et al., 2018; Kind et al., 2018) or non-binomial (Dittrich et al., 2019; Fletcher, Lickley, & Strzepek, 2019) learning scenarios generate larger state spaces.

2.2. Application of the Decision Framework to Lübeck

We apply our decision framework to the coastal city of Lübeck, Germany.

2.2.1. Case Study Description and Adaptation Options

The Hanseatic city of Lübeck is located on the German Baltic coast and has a population of 222,000 (Lübeck, 2023). The city lies at the mouth of the River Trave and its old town island, which is a UNESCO World





Figure 3. A map of Lübeck, Germany, showing different parts of the city, including the old town, Schlutup and Travemünde.

Heritage Site, lies 17 km inland from the estuary mouth (Figure 3). The city of Lübeck currently has no coastal flood protection infrastructure and is exposed to relatively frequent extreme events, such as the 1 in 10 years event. In October 2023, a storm surge with an extreme water level of 1.6 m flooded parts of the city of Lübeck, including the UNESCO World Heritage Site (NDR, 2023). Storm surges in Lübeck typically occur during strong westerly and northwesterly winds. Tides in the region are small (<10 cm) and thus contribute little to storm surge heights. As the Baltic Sea is a semi-enclosed basin, seiches can contribute several decimeters to the total water level. Therefore, water levels are affected by large-scale atmospheric forcing acting over the entire Baltic Sea (Jensen & Müller-Navarra, 2008).

Employees from the City of Lübeck participated in a stakeholder workshop to discuss preferences for the decision framework regarding adaptation options, confidence in sea level rise projections, and the time horizon. As a result, the majority of participants favored a longer time horizon of 100 years to take into account the lifetime of inflexible infrastructure. Further, the AR6 medium confidence sea level rise projections were preferred.

Three potential adaptation infrastructures were identified for the city of Lübeck: (i) a storm surge barrier in Travemünde with adjacent dikes; (ii) a storm surge barrier in Schlutup with adjacent dikes; and (iii) sea walls along the river in the old town, Schlutup and Travemünde (see Figure 4). We assume that only one of the three adaptation options will be implemented over time and that switching between the options is impossible. The decision-maker has to decide when and how high an initial adaptation action should be implemented (if any), and if and when further height upgrades should be implemented.

2.2.2. Sea Level Rise Learning Scenarios

No vertical land motion is projected for the city of Lübeck (Peltier et al., 2015), so the relative sea level rise is equal to the regional sea level rise. We use the medium confidence sea level rise projections from the AR6 p-box scenario for Travemünde (Fox-Kemper et al., 2021; Garner, Hermans, et al., 2021; Garner, Kopp, et al., 2021) to generate learning scenarios using the direct fit method of Völz and Hinkel (2023b). The AR6 regional p-box



Earth's Future

10.1029/2024EF004704



Figure 4. The three alternative adaptation options implemented in our framework: (i) sea walls along the river in the old town, Schlutup and Travemünde (top); (ii) a storm surge barrier in Travemünde with adjacent dikes (bottom left) and (iii) a storm surge barrier in Schlutup with adjacent dikes (bottom right).

scenarios combine probability distributions of several workflows into one scenario by taking the minimum quantile value of all workflow quantile values below 50%, the mean of all 50% quantile values, and the maximum quantile value of all workflow quantile values above 50% (Fox-Kemper et al., 2021; Garner, Hermans, et al., 2021; Garner, Kopp, et al., 2021). While the Monte Carlo samples of the AR6 sea level rise projections could also be used to generate learning scenarios, the use of p-box scenarios allows us to consider a wider range of uncertainty due to the alternative land ice emulators considered in the p-box scenarios, although this might overlook some other physical aspects. The resulting sea level rise learning scenarios for low and high climate change scenarios are shown in Figure 5.

2.2.3. Investment Costs

We model the initial investment costs (d = 0) for dikes and seawalls with a linear cost function depending on the length and height of a protection measure, as a regression analysis of Lenk et al. (2016) shows that linear cost functions are sufficient to model dike investment costs:

$$I(0, u) = c * u * b,$$
 (11)

where c is the unit cost factor (EUR/km*cm), u is the height of protection (cm) and b is the length of the coastal protection measure (km). Following an analysis by Mooyaart et al. (2014), we model the investment costs for storm surge barriers with a linear cost function depending only on the width of the barrier. We assume that a storm







Figure 5. Sea level rise learning scenarios for Travemünde generated by applying the direct fit method of Völz and Hinkel (2023b) to the AR6 medium confidence sea level rise projections. The learning scenarios are binomial lattices with a 50% probability of either rising or falling in each state. (a) shows a learning scenarios for SSP1-2.6 (low climate change) and (b) for SSP5-8.5 (high climate change).

surge barrier is always built with a protection height of 8 m and that, unlike dikes and sea walls, no further height upgrades are possible.

In line with the findings of Jonkman et al. (2013), we assume that subsequent height upgrades of dikes or seawalls are 1.5 times more expensive than the initial investment costs. We have further extended the linear cost function for height upgrades (d > 0) with a fixed cost component c_{fix} , to ensure that incremental dike upgrades are more costly than combining all incremental dike upgrades into an aggregated dike upgrade (Eijgenraam et al., 2017):

$$I(d, u) = c_{fix} * 1.5 * c * 1 * b + (1 - c_{fix}) * 1.5 * c * u * b, \quad d > 0.$$
(12)

We estimate the fixed cost component c_{fix} to be 30%, which is consistent with data from several Dutch ring dikes (Eijgenraam et al., 2017).

In our decision framework, two different types of dikes or seawalls can be built: either a normal inflexible protection measure, or a *flexible* protection measure (called a "climate dike" in Germany (Schwenn, 2021)), that has a wider or deeper foundation in anticipation of subsequent height upgrades in the future. The initial investment costs for *flexible* protection measures are 30% higher than for *normal* protection measures, but in return provide the opportunity of later, cheaper height upgrades up to twice the initial height. Thus, initial investment costs for flexible dikes or seawalls are assumed to be $1.3 \times I(0, u)$ and subsequent upgrade costs are assumed to be $0.5 \times I(d, u)$ for doubling the initial protection height.

We assume three different scenarios of low, medium or high unit costs based on other publications (J. Aerts, 2018; Hillen et al., 2010; Jonkman et al., 2013; Mooyaart et al., 2014; Nicholls et al., 2019; Prahl et al., 2018; Rehdanz et al., 2022). Table S3 in Supporting Information S1 shows the unit cost factors for dikes, sea walls and storm surge barriers, which we assume in this study. We estimate the maintenance costs for dikes, sea walls and barriers at 1% of the respective initial investment costs (J. Aerts, 2018; J. C. Aerts et al., 2013; Mooyaart et al., 2014) and the operating costs for barriers at 5% of the initial investment costs for barriers (Jonkman et al., 2013).

2.2.4. Flood Damage Modeling and Emulation

We use a hydrodynamic flood model developed by Kupfer et al. (2024) specifically designed for the city of Lübeck to model the flood propagation of storm surge heights. The hydrodynamic model takes into account the full water level hydrographs, which were developed using the method of MacPherson et al. (2019). The hydrodynamic flood model θ determines maximum flood depths w_p for each 10 × 10 m horizontal pixel p under storm surge height *x*, sea level rise *l* and adaptation action *d*:



$$\theta(l+x,d) \mapsto w_p \ \forall \ p \in P.$$
(13)

We apply a depth-damage function Δ to determine the relative asset damage a_p in pixel p under the maximum flood depth w_p aligned to Huizinga et al. (2016):

$$\Delta(w_p) \mapsto a_p. \tag{14}$$

Multiplying the relative damage a_p by the building value e_p gives the absolute damage $A_p^{x,l,d}$ for pixel p:

$$A_p^{x,l,d} = a_p * e_p = \Delta(\theta(l+x,d)) * e_p, \tag{15}$$

where buildings are identified using OpenStreetMaps and the building values are derived from official production cost tables published by the German Federal Ministry of Housing, Urban Development and Building. We provide a detailed explanation of the building value estimation process in Supporting Information S1. The total flood damage $A^{x,l,d}$ for the whole city of Lübeck, depending on storm surge height *x*, sea level rise *l* and adaptation action *d*, is determined by summing over all pixels:

$$A^{x,l,d} = \sum_{p} A_{p}^{x,l,d}.$$
 (16)

The flood simulation model has been applied to three different input data sets, one is a Latin Hypercube Sampling $(x \in [170, 340], l \in [0, 455], d \in [200, 800])$, one contains samples without adaptation and one contains sample points with adaptation below 200 cm. We combine all three data sets into one and apply oversampling to avoid data bias. Specifically, we define grid cells with increments of 200 cm over the total water level (l + x) and adaptation height (d) and duplicate the data points in each grid cell until all grid cells contain the same number of data points.

Our decision framework requires the total flood damage $A^{x,l,d}$ for more input points than given in our data set, so we mimic the flood simulation model, the pixel-wise application of the depth-damage function, and the aggregation of housing value damages (Equations 13–16) with a *k*-nearest-neighbor (*k*-NN) regression ϕ :

$$\phi(l+x,d) \mapsto A^{x,l,d},\tag{17}$$

with k = 2 and distance weighting, for example, giving more weight to points which are nearby. For the adaptation option *Schlutup barrier* and *sea walls* we use two *k*-NN regressors for pixels that are protected by the infrastructure and those that are not.

We assign probabilities to storm surge heights using a generalized extreme value (GEV) distribution provided by MacPherson et al. (2023). Rather than relying solely on high-resolution data (at least hourly samples), MacPherson et al. (2023) apply a Bayesian approach to extreme value analysis, which allows the incorporation of historical measurements in addition to the systematic tide-gauge data. Consequently, the resulting storm surge model incorporates information from a much longer observation period compared with traditional methods (Coles, 2001), providing benefits such as reduced estimate uncertainties and better representation of extraordinary events. This is particularly useful along the German Baltic Sea coast, which experienced very large events before the introduction of systematic tide-gauge records (Jensen et al., 2022). The final distribution function is based on 195 annual maxima sea levels, observed from 1826 to 2020, and includes several large historical events dating back to 1,044. The shape, scale and location parameters are -0.0175325, 0.28537, and 1.24056 respectively. We use this GEV distribution for f(x) in Equation 6. To make sure that we consider all possible extreme events, we set x_{max} to 8 m, which corresponds to an extreme event with a return frequency higher than a billion years.

2.2.5. Conducted Simulations

We apply our decision framework to the Lübeck case study and perform multiple runs of our non-adaptive and adaptive decision frameworks, varying the following parameters: low, medium or high unit costs; low climate

change (limit warming to 2°C; SSP1-2.6), medium climate change (limit warming to 3°C; SSP2-4.5) or high climate change (exceed warming of 4°C; SSP5-8.5); a low discount rate (1.4% as in Stern (2006)), a medium discount rate (2.0%) or a high discount rate (4.3% as applied by Nordhaus (2006) (Goulder & Williams, 2012)); and the Travemünde barrier, the Schlutup barrier or sea walls as adaptation options. This results in 81 simulation runs for each decision type (single-stage non-adaptive, multi-stage non-adaptive and adaptive) and 243 simulation runs in total. With the exception of the sensitivity analysis of the discount rates, we choose a constant discount rate of 1.4% for our results, which corresponds to the discount rates recommended by the German Central Bank (Bundesbank, 2023). We consider an adaptation state set consisting of incremental adaptation heights of 50 cm and a maximum height of 8 m. We consider 10-year time steps and assume that it takes one time step to implement adaptation actions.

3. Results

3.1. Comparison Between the Results of the Non-Adaptive and Adaptive Decision Analysis Methods

Comparing the results of single-stage non-adaptive, multi-stage non-adaptive and adaptive decision analysis methods under the medium unit cost assumption for the Travemünde barrier, we find that multi-stage non-adaptive and adaptive adaptation pathways recommend an initial protection height of 150 cm, irrespective of the climate change scenario (Figure 6). In contrast, the single-stage adaptation pathways recommend an initial protection height of 250 cm under high climate change. The multi-stage adaptation pathways recommend one future protection height upgrade under low climate change and two future protection height upgrades under high climate change. In contrast, the adaptive adaptation pathways recommend a wide range of later height upgrades that vary in timing and protection height and depend on future sea level rise thresholds.

Our results illustrate how adaptive adaptation pathways prevent over- and under-protection by recommending a wide range of cost-optimal adaptation actions, ranging from a low protection height of 1.5 m for very low sea level rise realizations (i.e., below 45 cm until 2150) to higher protection heights (e.g. 3 and 4 m) under higher sea level rise thresholds (e.g., above 90 cm until 2100 and above 160 cm until 2100) (Figures 6e and 6f). If a sea level rise threshold of 120 cm or 170 cm is reached before 2100, the adaptive adaptation pathway under high climate change for the Travemünde barrier results in an annual overtopping probability of 0.3% in 2100 (Table 1). In contrast, non-adaptive adaptation pathways can suffer greatly from under-protection if a high end sea level rise projection were to materialize (Figure 6). For example, reaching 120 cm or 170 cm of sea level rise before 2100 would result in annual overtopping probabilities in 2100 of 55.6% or 99.0% (Table 1).

The single-stage non-adaptive decision analysis method overestimates adaptation costs compared to multi-stage non-adaptive decision analysis methods, and adaptive decision analysis methods have the potential to reduce adaptation costs further (Table 2). Learning about sea level rise uncertainty, implemented in the adaptive adaptation pathway, reduces the overall expected costs of the Travemünde barrier under the medium unit cost assumption by 70 million euros (-1.8%) compared to the multi-stage adaptation pathway (Table 2). We call this cost reduction stemming from using sea level rise observations to update future sea level rise projection the value of learning. In prevalent literature this term is also called the value of information, value of flexibility or option value (Dawson et al., 2018; Kwakkel, 2020; Liu et al., 2018; Pachos et al., 2022; Skerker et al., 2023; Wreford et al., 2020). The value of learning is 29 million euro (-0.5%) for the Schlutup barrier and 41 million euro (-0.9%) for the sea walls respectively. For all adaptation options the value of multi-stage decision-making (single-stage vs. multi-stage adaptation pathways) is higher than the value of learning (Table 2). The difference in the value of learning and multi-stage decision-making between the adaptation options can be explained by the costs of height upgrades of the different options. The more costly it is to implement or upgrade the protection heights, the more beneficial it is to either postpone the protection height upgrades into the future or to learn about sea level rise in order to avoid unnecessary protection height upgrades. The upgrade costs for the Travemünde barrier are the highest, because the complementary dikes of the Travemünde barrier are more than twice as long as the complementary dikes of the Schlutup barrier (Figure 4) and the unit costs for dikes are higher than for sea walls (Table S3 in Supporting Information S1). Consequently, the value of multi-stage decision-making and the value of learning are significantly higher for the Travemünde barrier than for the Schlutup barrier or the sea walls.



Earth's Future

a) Single- stage method for SSP1- 2.6





Figure 6.

3.2. Sensitivity Analyses for Adaptive Adaptation Pathways

On average, the Travemünde barrier is the most efficient adaptation option for the city of Lübeck and the sea walls are the least efficient adaptation option (Figure 7), which can be explained by the effectiveness of the Travemünde barrier in reducing flood damage due to its location at the mouth of the river. Adaptation actions are always efficient today for low or medium unit costs (Figure 9). Only in the unlikely situation of high unit costs and low end sea level rise, no adaptation action is recommended at all over the whole planning horizon (Figure 9). Sea walls have a wider range of initial protection heights from 0.5 to 3.5 m compared to barriers with 1.5-2 m (Figure 7), as this range depends on the unit cost assumption (Figure 9) and sea walls have significantly lower unit costs (Table S3 in Supporting Information S1), high initial protection heights are efficient at low unit costs and vice versa. Implementing the flexible adaptation option with a wider or deeper foundation for subsequent height upgrades is only efficient for a minority of adaptive simulation runs, namely for barriers with high unit costs under high climate change (Figures 7-9). Higher climate change increases expected flood damages and investments, but does barely affect the earliest action parameters (Figure 8). High unit costs, in contrast to low or medium unit costs, have a major impact on adaptive adaptation pathways; not only do they greatly increase overall costs, but they also lead to delays in early action and prevent adaptation options from being efficient at low end sea level rise (Figure 9). Higher discount rates devalue future capital flows and significantly reduce flood damages and investment costs, while the mean parameters of the earliest adaptation actions are almost unaffected by different discount rates (Figure S3 in Supporting Information S1).

The climate change scenarios have almost no effect on the earliest adaptation actions, only the maximum values are different when comparing the low and medium climate change scenario (Figure 8). In particular, the adaptive adaptation pathways in Figures 6e and 6f for low and high climate change scenarios recommend similar protection heights for similar sea level rise thresholds. For example, both pathways recommend an initial protection height of 1.5 m and no further protection height upgrade if sea level rise stays below 45 cm. If a sea level rise threshold of 90 cm is reached before 2100, both adaptive adaptation pathways recommend a protection height upgrade to 3 m. Similar, if a sea level rise threshold of 160 cm is reached before 2100, both pathways recommend a protection height upgrade to 4 m. This detailed example for the Travemünde barrier (Figures 6e and 6f), together with the sensitivity analysis in Figure 8, shows that adaptive adaptation pathways generate sea level rise thresholds for adaptation actions that are similar across climate change scenarios, but pathways under high climate change consider higher high-end sea level rise. In contrast, non-adaptive adaptation pathways for different climate change scenarios (Figures 6a–6d).

Our results indicate that, from an economic perspective, the city of Lübeck is currently under-protected against storm surges and that adaptation actions should be implemented today. Based on our findings, an adaptation strategy for the city of Lübeck could be to implement a storm surge barrier and adjacent dikes in Travemünde and to monitor future sea level rise to trigger height upgrades. Once a certain threshold of sea level rise is observed, the adjacent dikes should be upgraded, with the exact height of the upgrade depending on the projected unit costs. A barrier option also has the advantage over sea walls that the old town island of Lübeck will not lose its status as a UNESCO World Heritage Site.

4. Discussion

Our results show that, from an economic perspective, adaptation actions should be implemented today in the city of Lübeck, in line with the call for immediate adaptation actions in AR6 (Cooley et al., 2022). Most economic assessments of coastal adaptation have considered non-adaptive decision analysis methods and thus ignored the timing of adaptation actions, that is, assumed that they need to be taken today. Adaptive adaptation pathways derived from real-option analysis or dynamic planning studies can quantify the optimal timing of adaptation

Figure 6. Adaptation pathways resulting from the applications of the single-stage non-adaptive (a, b), multi-stage non-adaptive (c, d) and adaptive (e, f) decision analysis methods for the Travemünde barrier under the assumption of medium unit costs and a discount rate of 1.4% for low climate change on the left (SSP1-2.6, 2° C) and high climate change on the right (SSP5-5.8, 4° C). For the adaptive adaptation pathways (e, f), we indicate the thresholds of observations of sea level rise *l* (cm) at which adaptation actions are triggered. For example, in the adaptive adaptation pathway for low climate change (e), observing a sea level rise of 59 cm or higher in 2070 triggers a protection height upgrade of 50 cm, that is, transferring from the blue to the yellow adaptation pathway.



Table 1

Annual Overtopping Probability P for the Year 2100 Under Optimal Protection Heights for the Schlutup Barrier, Travemünde Barrier and Sea Walls Under High Climate Change (SSP5-8.5), Medium Unit Costs and a Discount Rate of 1.4% for the Single-Stage Non-Adaptive, Multi-Stage Non-Adaptive and Adaptive Decision Analysis Method

	l (SLR) (cm)	Single-stage		Multi-stage		Adaptive	
Adaptation option		d (cm)	$\mathcal{P}\left(\% ight)$	d (cm)	$\mathcal{P}\left(\% ight)$	d (cm)	$\mathcal{P}\left(\% ight)$
Schlutup barrier	40	250	4.4	250	4.4	150	80.5
	80	250	17.7	250	17.7	200	68.4
	120	250	55.6	250	55.6	300	12.7
	170	250	99.0	250	99.0	450	0.3
Travemünde barrier	40	250	4.4	250	4.4	150	80.5
	80	250	17.7	250	17.7	300	3.1
	120	250	55.6	250	55.6	400	0.3
	170	250	99.0	250	99.0	450	0.3
Sea walls	40	350	0.1	400	0.0	300	0.7
	80	350	0.5	400	0.1	300	3.1
	120	350	3.1	400	0.3	400	0.3
	170	350	12.7	400	2.1	400	2.1

Note. We assess four different sea level rise values l that will be reached before 2100 to determine the exact overtopping probability. We determine the annual overtopping probability by integrating the GEV function describing the probabilities of storm surge heights (Section 2.2.4) and set the lower bound of the integration to the optimal protection height d (Figure 6) minus the sea level rise value. Note that for the multi-stage adaptation pathway of the Travemünde barrier a protection height upgrade is triggered in 2100, but the current protection level d in 2100 is still 250 cm (Figure 6).

actions and shed light on the urgency of coastal adaptation actions from an economic perspective (Ginbo et al., 2020; Wreford et al., 2020).

The adaptive adaptation pathways for the city of Lübeck are similar across climate change scenarios, showing that scenario-based application for each climate change scenario (Oppenheimer et al., 2019) and comparison of the adaptive adaptation pathways can address the problem of deep uncertainty in the case of sea level rise (i.e., it is not possible to assign probabilities to future scenarios). A major criticism of real-option analysis and other economic decision-making methods based on expected values is that they are impractical when there is deep uncertainty, as is the case with climate change scenarios (Kwakkel, 2020; Oppenheimer et al., 2019). We have shown that adaptive adaptation pathways can be similar across climate change scenarios, that is, the sea level rise thresholds and triggered protection heights of the adaptive adaptation pathway for low climate change and high climate change are similar, and one can follow the adaptive adaptation pathways are not similar across climate change scenarios pathways are not similar across climate change scenarios pathways for high climate change to account for potential high-end sea level rise. In contrast, the non-adaptive adaptation pathways are not similar across climate change scenarios in our results, and are indeed impractical in settings with deep uncertainty.

The values of learning in our study (Table 2) indicate that learning about sea level rise uncertainty is valuable for coastal decision settings with planning horizons beyond 2050, high investment costs of changing adaptation

Table 2

Today's Expected Costs (V_{2020} From the Bellman Equation 8) of All Future Flood Damages and Adaptation Investments for the Schlutup Barrier, Travemünde Barrier and Sea Walls Under High Climate Change (SSP5-8.5), Medium Unit Costs and a Discount Rate of 1.4% for the Single-Stage Non-Adaptive, Multi-Stage Non-Adaptive and Adaptive Decision Analysis Method

	Single-stage (Mio. EUR)	Multi-stage (Mio. EUR)	Adaptive (Mio. EUR)	VoM (Mio. EUR)	VoL (Mio. EUR)
Schlutup barrier	5,217	5,095	5,066	122	29
Travemünde barrier	4,347	3,992	3,922	355	70
Sea walls	4,888	4,675	4,634	213	41

Note. We define the *value of multi-stage decision-making* (VoM) as the difference between the single-stage non-adaptive and the multi-stage non-adaptive decision analysis method and the *value of learning* (VoL) as the difference between the multi-stage non-adaptive and the adaptive decision analysis method.



Earth's Future

10.1029/2024EF004704



Figure 7. Comparison of adaptive adaptation pathways for the Travemünde barrier, the Schlutup barrier and sea walls. The statistics shown consider low, medium and high unit costs and low (SSP1-2.6), medium (SSP2-4.5) and high (SSP5-8.5) climate change. In the first row we present statistics on the height of protection, year and sea level rise value of the earliest possible initial investment of each adaptive adaptation pathway. The second row shows the average total costs, the ratio of adaptive adaptation pathways that use flexible options, and the ratio of adaptive adaptation pathways that apply adaptation even if there is no sea level rise (the lowest sea level rise path in the learning scenario is realized).



Figure 8. Comparison of adaptive adaptation pathways for the low (SSP1-2.6), medium (SSP2-4.5) and high (SSP5-8.5) climate change scenario. The statistics shown consider low, medium and high unit costs and the three different adaptation options (Travemünde barrier, Schlutup barrier and sea walls). In the first row we present statistics on the height of protection, year and sea level rise value of the earliest possible initial investment of each adaptive adaptation pathway. The second row shows the average total costs, the ratio of adaptive adaptation pathways that use flexible options, and the ratio of adaptive adaptation pathways that apply adaptation even if there is no sea level rise (the lowest sea level rise path in the learning scenario is realized).



Figure 9. Comparison of adaptive adaptation pathways for a low, medium and high unit costs as indicated in Table 2. The statistics shown consider low (SSP1-2.6), medium (SSP2-4.5) and high (SSP5-8.5) climate change, the three different adaptation options (Travemünde barrier, Schlutup barrier and sea walls) and a discount rate of 1.4%. In the first row we present statistics on the height of protection, year and sea level rise value of the earliest possible initial investment of each adaptive adaptation pathway. The second row shows the average total costs, the ratio of adaptive adaptation pathways that use flexible options, and the ratio of adaptive adaptation pathways that apply adaptation even if there is no sea level rise (the lowest sea level rise path in the learning scenario is realized).

pathways, and/or no need for adaptation action until the middle of this century. Our results indicate that significant learning about sea level rise uncertainty takes place beyond 2050, as this is not only when the earliest height upgrades are recommended in our adaptive adaptation pathways (Figure 6), but also when the uncertainty in sea level rise projections increases significantly (Fox-Kemper et al., 2021). As it is inefficient to delay the initial adaptation action in our case study, the value of learning is higher for adaptation options with higher costs of changing adaptation pathways, as future learning can avoid unnecessary protection height upgrades, as in the case of the Travemünde barrier (Table 2). We also expect a high value of learning in decision settings where adaptation actions can be delayed until the middle of this century, as the initial adaptation can then be adjusted to sea level rise observations and respective learning. Similarly, adaptive pathway analysis and real-option analysis are particularly valuable in decision contexts involving long-term decisions, costly investments, uncertainty, and expected future learning opportunities (Haasnoot et al., 2020; Wreford et al., 2020).

Applying the flexible adaptation option for sea walls or complementary dikes next to storm surge barriers is inefficient in most of our simulation runs. Vogelsang et al. (2023) find that flexible adaptation options are particularly valuable when there is a fast learning rate, but most of the learning about sea level rise uncertainty takes place after 2050. Because coastal protection is needed today in our study, but there is no fast speed of learning until 2050, building flexible adaptation options today is often inefficient in our simulation runs. Similar to Skerker et al. (2023), we also find that the use of the flexible adaptation option is highly depended on the discount rate. We find that at a discount rate of 1.4%, a flexible adaptation option is only efficient in 7% of our simulation runs, but at a discount rate of 0%, a flexible adaptation option is efficient in 48% of our simulation runs. As the discount rate devalues future capital flows, it also reduces the efficiency of flexible adaptation options that involve higher expenses today in return for future cost savings.

In our application, the main uncertainty in adaptive adaptation pathways comes from unit costs, but these can be resolved once concrete planning of adaptation actions begins. The sensitivity analysis of unit costs shows the largest differences between unit costs compared to the sensitivity of the discount rate, climate change and adaptation options (Figures 7–9 and Figure S3 in Supporting Information S1). Once specific unit costs are known, the results can be filtered to show only results for the appropriate unit cost.

Non-adaptive decision analysis methods that ignore future learning through observations, although widely used in normative analysis of coastal adaptation decisions (Bachner et al., 2022; Eijgenraam et al., 2014, 2017; Lincke & Hinkel, 2018; Tiggeloven et al., 2020; van Dantzig, 1956; Van der Pijl & Oosterlee, 2012; Vousdoukas et al., 2020), are less likely to represent real-world decisions than adaptive decision analysis methods that take into account future learning. It is unrealistic to assume that decision-makers would implement adaptation measures designed a century ago without revision. In fact, recent literature on coastal adaptation to climate change strongly recommends iterative decision-making and frequent reassessment of actions (Hinkel et al., 2019; Oppenheimer et al., 2019).

By presenting optimal coastal adaptive adaptation pathways for the next 130 years, we do not mean to recommend following these pathways for the next century without revising them. Rather, these results serve, similarly to any other result of normative decision analysis over time, as a valuable basis for informing choices in the short term. But even for this, it is important to include the long planning horizon in the economic assessment, as the flexibility to change course and learn in the long-term also influences optimal decisions in the short term (Zwaneveld et al., 2018), for example, by reducing today's expected costs (Table 2). The multi-stage non-adaptive coastal adaptation model of Eijgenraam et al. (2017), which has been applied for finding optimal protection levels in the Delta Program 2015 (the official coastal protection guideline of the Netherlands), considers a time horizon of 300 years. The Dutch government plans to follow the results of Eijgenraam et al. (2017) in the short term, that is, the next 20 or 30 years, and then revise the strategies. In general, it is always advisable to monitor sea level rise and update adaptation strategies over time, regardless of the method used to develop the original strategy (Oppenheimer et al., 2019). As this is a relatively new field of research, there is little experience of the real-world challenges of applying such a framework. Apart from Eijgenraam et al. (2017), we are not aware of any other examples of coastal adaptation where multi-stage non-adaptive or adaptive decision analysis methods from economic assessment studies have been used. Given the rather technical and complex nature of the methodology, we note that well-trained staff and sufficient resources are required to carry out such an analysis.

A limitation of our model is that we only consider flood damage to buildings and ignore flood damage to infrastructure, such as the sewage treatment plant in Travemünde, and cascading effects, such as lost revenue from closed businesses. Previous studies have addressed infrastructure damages and cascading effects of flood events and found that the impacts of floods can be much broader than their direct effects, but economic evaluations are scarce (Fekete, 2019, 2020; Hilly et al., 2018; Unterberger, 2018). While our underlying hydrodynamic model results can be used to analyse flood hazards to critical infrastructure, the inclusion of critical infrastructure and cascading effects in the economic evaluation is difficult due to data scarcity. For example, the data source we use to estimate the value of buildings in Germany does not include information on the value of wastewater treatment plants. As we only consider direct flood damage to buildings, we have a rather conservative estimate of annual flood damage in our study.

Another limitation of our study is that we assume static building values over time, whereas in reality buildings are likely to increase over time with socio-economic development in the floodplain. This assumption of our study, similar to the one discussed in the previous paragraph, also leads to a rather conservative estimate of flood damage. A follow-up study could include socio-economic development over time and explore the effect of different pathways of socio-economic development. Considering future learning about socio-economic development in combination with future learning about sea level rise would provide interesting insights into the interdependencies between exposure and hazard and their effect on efficient adaptation strategies, which have so far only been analysed with highly simplified quantifications for future learning (Völz & Hinkel, 2023a).

We assume a stationary extreme value distribution to model storm surge heights in our decision framework, which does not incorporate potential changes in storm surge heights due to the effects of climate change. To date, model results along the German coastline have been inconsistent, and no conclusions could be drawn about the effects of climate change on storm surge heights. For example, Gräwe and Burchard (2012) considered the non-linear effects of sea level rise on future storm surges at the western Baltic Sea and found that, at the location of our case study, increases to future storm surges are equivalent to changes in sea level rise. However, according to Vousdoukas et al. (2017), the Baltic Sea region is expected to experience large changes in the height of future storm surges. While it is likely that the distribution of storm surge heights will change in the future, it is unclear how this will unfold, and further insights are needed.

23284277, 2024, 10, Downloaded from https

onlinelibrary.wiley.com/doi/10.1029/2024EF004704 by Cochrane France

, Wiley Online Library on [04/02/2025]. See the Terms

and Conditic

ns) on Wiley Online Library for rules

are go

med by the applicable Creative

Our hydrodynamic flood model limits our choice of adaptation options because it is based on 10×10 m grid cells. This allows us to model large infrastructure protection, but not smaller adaptation options such as mobile flood barriers or nature-based adaptation such as salt marshes. We note that alternative adaptation options such as mobile flood barriers, dry- or wet-proofing of buildings and nature-based solutions could be alternatives or extensions to the adaptation options proposed in our paper. Exploring the efficiency of these alternative adaptation options would be an interesting follow-up study. For example, consideration of dry- or wet-proofing could be done through specific depth-damage functions, as empirically developed by Endendijk et al. (2023). Recent work in the context of river floods and droughts by Vogelsang et al. (2023) applies real-option analysis to consider retention areas as a nature-based adaptation option with respect to learning about climate change. To date, there are no economic studies of adaptive decision-making that consider any of the above alternative adaptation options in combination with future learning about sea level rise uncertainty (Völz & Hinkel, 2023a).

Alternative adaptive decision analysis methods, such as online optimization, policy search, Bayesian approaches and reinforcement learning, could be explored to advance economic adaptive adaptation pathways for dealing with sea level rise. In addition, comparing the performance of these normative adaptive pathways with descriptive real-world decision-making practices can provide valuable insights. One could model real-world coastal decisionmaking practices, such as incrementally increasing dike heights in response to observed sea level rise, or taking immediate risk-averse adaptation actions to prepare for high-end sea level rise scenarios. By comparing the performance of these descriptive approaches with normative adaptation pathways, the effectiveness and efficiency of different strategies in managing sea level rise can be evaluated.

5. Conclusion

In this paper, we present an adaptive decision framework for developing efficient coastal adaptation, taking into account future learning about sea level rise uncertainty. We formulate a stochastic optimization problem and integrate a quantification of the opportunity to learn about future sea level rise uncertainty through learning scenarios. We apply this framework to a coastal case study in Germany and use hydrodynamic flood modeling to estimate annual flood damages. We find three main advantages of adaptive decision analysis methods over nonadaptive decision analysis methods that do not take into account future learning about sea level rise uncertainty. First, adaptive adaptation pathways generate sea level rise thresholds with corresponding optimal protection heights that can be similar across climate change scenarios. Second, adaptive adaptation pathways prevent costly over- or under-protection against flood risks by taking into account future observations of sea level rise, for example, two protection height upgrades are recommended for the Travemünde barrier option if sea level rises above 1.2 m before 2100, resulting in an annual overtopping probability of 0.3%, compared to 56% for nonadaptive methods. Third, adaptive adaptation pathways reduce the expected costs of adaptation investments and flood damages compared to non-adaptive pathways, in our study by up to 1.8%.

Data Availability Statement

The data and code used in this study are publicly available in Völz et al. (2024) under https://zenodo.org/doi/10. 5281/zenodo.10262546.

References

Aerts, J. (2018). A review of cost estimates for flood adaptation. Water, 10(11), 1646. (Publisher: MDPI AG tex.readstatus: skimmed). https://doi. org/10.3390/w10111646

Aerts, J. C., Botzen, W. W., De Moel, H., & Bowman, M. (2013). Cost estimates for flood resilience and protection strategies in New York City. Annals of the New York Academy of Sciences, 1294(1), 1–104. https://doi.org/10.1111/nyas.12200

Anthoff, D., Nicholls, R. J., & Tol, R. S. J. (2010). The economic impact of substantial sea-level rise. Mitigation and Adaptation Strategies for Global Change, 15(4), 321-335. https://doi.org/10.1007/s11027-010-9220-7

Bachner, G., Lincke, D., & Hinkel, J. (2022). The macroeconomic effects of adapting to high-end sea-level rise via protection and migration. Nature Communications, 13(1), 5705. https://doi.org/10.1038/s41467-022-33043-z

Bellman, R. (1966). In Dynamic programming (Vol. 153). Bundesbank, D. (2023). Discount interest rates pursuant to section 253 (2) of the German Commercial Code/7-year average. Retrieved from https:// www.bundesbank.de/en/statistics/money-and-capital-markets/interest-rates-and-yields/discount-interest-rates/discount-interest-rates/2002 and a statistics/money-and-capital-markets/interest-rates-and-yields/discount-interest-rates/discount-interest-rates/2002 and a statistics/money-and-capital-markets/interest-rates-and-yields/discount-interest-rates/discount-interest-rates/discount-interest-rates/2002 and a statistics/money-and-capital-markets/interest-rates/and-yields/discount-interest-rates/discount-i

Coles, S. (2001). An introduction to statistical modeling of extreme values. Springer. Cooley, D. S., Schoeman, L., Bopp, P., Boyd, S., Donner, D., Ghebrehiwet, S.-I., et al. (2022). Oceans and coastal ecosystems and their services. In H.-O. Pörtner, D. C. Roberts, M. Tignor, E. S. poloczanska, K. Mintenbeck, A. Alegría, et al. (Eds.), Climate Change 2022: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 379-550). Cambridge University Press. https://doi.org/10.1017/9781009325844.005

This publication has been funded by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) within the Special Priority Program SPP-1889 SeaLevel-project SEASCApe II. This publication was supported by PROTECT. This project has received funding from the European Union's Horizon 2020 research and innovation programme under Grant 869304, PROTECT contribution number 132. This project has received funding from the European Union's Horizon 2020 research and innovation programme CoCliCo under Grant 101003598. Additional methods used in this study are provided in Supporting Information S1. There are no conflicts of interests to declare. We thank the projection authors for developing and making the sea level rise projections available, multiple funding agencies for supporting the development of the projections, and the NASA Sea Level Change Team for developing and hosting the IPCC AR6 Sea Level Projection Tool. We acknowledge support by the Open Access Publication Fund of Humboldt-Universität zu Berlin. Open Access funding enabled and organized by Projekt DEAL.

- Dawson, D. A., Hunt, A., Shaw, J., & Gehrels, W. R. (2018). The economic value of climate information in adaptation decisions: Learning in the sea-level rise and coastal infrastructure context. *Ecological Economics*, 150, 1–10. (Publisher: Elsevier tex.readstatus: read). https://doi.org/10. 1016/j.ecolecon.2018.03.027
- Diaz, D. B. (2016). Estimating global damages from sea level rise with the Coastal Impact and Adaptation Model (CIAM). *Climatic Change*, 137(1-2), 143–156. https://doi.org/10.1007/s10584-016-1675-4
- Dittrich, R., Butler, A., Ball, T., Wreford, A., & Moran, D. (2019). Making real options analysis more accessible for climate change adaptation. An application to afforestation as a flood management measure in the Scottish Borders. *Journal of Environmental Management*, 245, 338–347. (Publisher: Elsevier BV). https://doi.org/10.1016/j.jenvman.2019.05.077
- Eijgenraam, C., Brekelmans, R., Hertog, D. d., & Roos, K. (2017). Optimal strategies for flood prevention. *Management Science*, 63(5), 1644–1656. (Publisher: Institute for Operations Research and the Management Sciences (INFORMS)). https://doi.org/10.1287/mnsc.2015.2395
- Eijgenraam, C., Kind, J., Bak, C., Brekelmans, R., Den Hertog, D., Duits, M., et al. (2014). Economically efficient standards to protect The Netherlands against flooding. *Interfaces*, 44(1), 7–21. https://doi.org/10.1287/inte.2013.0721
- Endendijk, T., Botzen, W. J. W., De Moel, H., Aerts, J. C. J. H., Slager, K., & Kok, M. (2023). Flood vulnerability models and household flood damage mitigation measures: An econometric analysis of survey data. *Water Resources Research*, 59(8), e2022WR034192. https://doi.org/10. 1029/2022WR034192
- Erfani, T., Pachos, K., & Harou, J. J. (2018). Real-options water supply planning: Multistage scenario trees for adaptive and flexible capacity expansion under probabilistic climate change uncertainty. *Water Resources Research*, 54(7), 5069–5087. (Publisher: American Geophysical Union (AGU) tex.readstatus: read). https://doi.org/10.1029/2017wr021803
- Fekete, A. (2019). Critical infrastructure and flood resilience: Cascading effects beyond water. WIREs Water, 6(5), e1370. https://doi.org/10.1002/ wat2.1370
- Fekete, A. (2020). Critical infrastructure cascading effects. Disaster resilience assessment for floods affecting city of Cologne and Rhein-Erft-Kreis. Journal of Flood Risk Management, 13(2), e312600. https://doi.org/10.1111/jfr3.12600
- Fletcher, S., Lickley, M., & Strzepek, K. (2019). Learning about climate change uncertainty enables flexible water infrastructure planning. *Nature Communications*, 10(1), 1782. (Publisher: Springer Science and Business Media LLC). https://doi.org/10.1038/s41467-019-09677-x
- Fletcher, S., Strzepek, K., Alsaati, A., & de Weck, O. (2019). Learning and flexibility for water supply infrastructure planning under groundwater resource uncertainty. *Environmental Research Letters*, 14(11), 114022. (Publisher: IOP Publishing). https://doi.org/10.1088/1748-9326/ ab4664
- Fox-Kemper, B., Hewitt, H. T., Xiao, C., Aalgeirsdóttir, G., Drijfhout, S. S., Ed-wards, T. L., et al. (2021). Ocean, cryosphere and sea level change. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. connors, C. Péan, S. Berger, et al. (Eds.), *Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change* (pp. 1211–1362). Cambridge University Press, Cambridge. https://doi.org/10.1017/9781009157896.011
- Garner, G. G., Hermans, T., Kopp, R. E., Slangen, A. B. A., Edwards, T. L., Levermann, A., et al. (2021). IPCC AR6 Sea-Level Rise Projections. Version 20210809 [Dataset]. PO.DAAC, CA, USA. https://podaac.jpl.nasa.gov/announcements/2021-08-09-Sea-level-projections-from-the-IPCC-6th-Assessment-Report
- Garner, G. G., & Keller, K. (2018). Using direct policy search to identify robust strategies in adapting to uncertain sea-level rise and storm surge. Environmental Modelling and Software, 107, 96–104. https://doi.org/10.1016/j.envsoft.2018.05.006
- Garner, G. G., Kopp, R. E., Hermans, T., Slangen, A. B. A., Koubbe, G., Turilli, M., et al. (2021). Framework for assessing changes to sea-level (FACTS). *Geoscientific Model Development*, *16*(24), 7461–7489. https://doi.org/10.5194/gmd-16-7461-2023
- Gersonius, B., Ashley, R., Pathirana, A., & Zevenbergen, C. (2013). Climate change uncertainty: Building flexibility into water and flood risk infrastructure. *Climatic Change*, 116(2), 411–423. (Publisher: Springer Science and Business Media LLC tex.readstatus: read). https://doi.org/ 10.1007/s10584-012-0494-5
- Gersonius, B., Morselt, T., van Nieuwenhuijzen, L., Ashley, R., & Zevenbergen, C. (2012). How the failure to account for flexibility in the economic analysis of flood risk and coastal management strategies can result in maladaptive decisions. *Journal of Waterway, Port, Coastal, and Ocean Engineering, 138*(5), 386–393. (Publisher: American Society of Civil Engineers (ASCE)). https://doi.org/10.1061/(asce)ww.1943-5460.0000142
- Ginbo, T., Corato, L. D., & Hoffmann, R. (2020). Investing in climate change adaptation and mitigation: A methodological review of real-options studies. Ambio, 50(1), 229–241. (Publisher: Springer Science and Business Media LLC tex.ranking: rank5 tex.readstatus: read). https://doi.org/ 10.1007/s13280-020-01342-8
- Goulder, L. H., & Williams, R. C. W. (2012). The choice of discount rate for climate change policy evaluation. *Climate Change Economics*, 03(04), 1250024. https://doi.org/10.1142/S2010007812500248
- Gräwe, U., & Burchard, H. (2012). Storm surges in the Western Baltic Sea: The present and a possible future. *Climate Dynamics*, 39(1–2), 165–183. https://doi.org/10.1007/s00382-011-1185-z
- Haasnoot, M., Kwakkel, J. H., Walker, W. E., & ter Maat, J. (2013). Dynamic adaptive policy pathways: A method for crafting robust decisions for a deeply uncertain world. *Global Environmental Change*, 23(2), 485–498. (Publisher: Elsevier BV tex.readstatus: read). https://doi.org/10. 1016/j.gloenvcha.2012.12.006
- Haasnoot, M., Middelkoop, H., Offermans, A., van Beek, E., & van Deursen, W. P. A. (2012). Exploring pathways for sustainable water management in river deltas in a changing environment. *Climatic Change*, 115(3–4), 795–819. (Publisher: Springer Science and Business Media LLC). https://doi.org/10.1007/s10584-012-0444-2
- Haasnoot, M., Van Aalst, M., Rozenberg, J., Dominique, K., Matthews, J., Bouwer, L. M., et al. (2020). Investments under non-stationarity: Economic evaluation of adaptation pathways. *Climatic Change*, *161*(3), 451–463. https://doi.org/10.1007/s10584-019-02409-6
- Hallegatte, S., Green, C., Nicholls, R. J., & Corfee-Morlot, J. (2013). Future flood losses in major coastal cities. *Nature Climate Change*, *3*(9), 802–806. https://doi.org/10.1038/nclimate1979
- Herman, J. D., Quinn, J. D., Steinschneider, S., Giuliani, M., & Fletcher, S. (2020). Climate adaptation as a control problem: Review and perspectives on dynamic water resources planning under uncertainty. *Water Resources Research*, 56(2). (Publisher: American Geophysical Union (AGU)). https://doi.org/10.1029/2019wr025502
- Hillen, J., Kanning, K., & Geldenhuys, S. (2010). Coastal defence cost estimates. Communications on Hydraulic and Geotechnical Engineering. Hilly, G., Vojinovic, Z., Weesakul, S., Sanchez, A., Hoang, D., Djordjevic, S., et al. (2018). Methodological framework for analysing cascading effects from flood events: The case of Sukhumvit Area, Bangkok, Thailand. Water, 10(1), 81. https://doi.org/10.3390/w10010081
- Hinkel, J., Church, J. A., Gregory, J. M., Lambert, E., Cozannet, G. L., Lowe, J., et al. (2019). Meeting user needs for sea level rise information: A decision analysis perspective. *Earth's Future*, 7(3), 320–337. (Publisher: American Geophysical Union (AGU) tex.readstatus: read). https://doi. org/10.1029/2018ef001071

- Hinkel, J., Lincke, D., Vafeidis, A. T., Perrette, M., Nicholls, R. J., Tol, R. S. J., et al. (2014). Coastal flood damage and adaptation costs under 21st century sea-level rise. *Proceedings of the National Academy of Sciences of the United States of America*, 111(9), 3292–3297. https://doi.org/10. 1073/pnas.1222469111
- Huizinga, J., de Moel, H., & Szewczyk, W. (2016). Global flood depth-damage functions: Methodology and the database with guidelines. LU: Publications Office. https://doi.org/10.2760/16510
- Jensen, J., Mai, H., & Beckmann, S. (2022). Best estimates for historical storm surge water level and MSL development at the Travemünde/Baltic Sea gauge over the last 1,000 years (p. 27). (Artwork Size: 27 pages ISBN: 9783939230830 Medium: application/pdf/a Publisher: Kuratorium für Forschung im Küsteningenieurwesen (Hrsg.), via Bundesanstalt für Wasserbau). https://doi.org/10.18171/1.092102
- Jensen, J., & Müller-Navarra, S. H. (2008). Storm surges on the German Coast. In *Die Küste* (Vol. 74, pp. 92–124).
- Jonkman, S. N., Hillen, M. M., Nicholls, R. J., Kanning, W., & van Ledden, M. (2013). Costs of adapting coastal defences to sea-level rise—New estimates and their implications. *Journal of Coastal Research*, 290, 1212–1226. (Publisher: Coastal Education and Research Foundation). https://doi.org/10.2112/jcoastres-d-12-00230.1
- Kind, J. M., Baayen, J. H., & Botzen, W. J. W. (2018). Benefits and limitations of real options analysis for the practice of river flood risk management. Water Resources Research, 54(4), 3018–3036. (Publisher: American Geophysical Union (AGU)). https://doi.org/10.1002/ 2017wr022402
- Kupfer, S., MacPherson, L. R., Hinkel, J., Arns, A., & Vafeidis, A. T. (2024). A comprehensive probabilistic flood assessment accounting for hydrograph variability of ESL events. *Journal of Geophysical Research: Oceans*, 129(1), e2023JC019886. https://doi.org/10.1029/ 2023JC019886
- Kwakkel, J. H. (2020). Is real options analysis fit for purpose in supporting climate adaptation planning and decision-making? WIREs Climate Change, 11(3). (Publisher: Wiley). https://doi.org/10.1002/wcc.638
- Lenk, S., Rybski, D., Heidrich, O., Dawson, R. J., & Kropp, J. P. (2016). Costs of sea dikes—Regressions and uncertainty estimates. (Publisher: Copernicus GmbH). https://doi.org/10.5194/nhess-2016-270
- Lickley, M. J., Lin, N., & Jacoby, H. D. (2014). Analysis of coastal protection under rising flood risk. Climate Risk Management, 6, 18–26. https:// doi.org/10.1016/j.crm.2015.01.001
- Lincke, D., & Hinkel, J. (2018). Economically robust protection against 21st century sea-level rise. Global Environmental Change, 51, 67–73. https://doi.org/10.1016/j.gloenvcha.2018.05.003
- Lincke, D., & Hinkel, J. (2021). Coastal migration due to 21st century sea-level rise. Earth's Future, 9(5), e2020EF001965. https://doi.org/10. 1029/2020EF001965
- Linquiti, P., & Vonortas, N. (2012). The value of flexibility in adapting to climate change: A real options analysis of investments in coastal defense. Climate Change Economics, 3(02), 1250008. (Publisher: World Scientific tex.readstatus: read). https://doi.org/10.1142/ s201000781250008x
- Liu, H., Wang, Y., Zhang, C., Chen, A. S., & Fu, G. (2018). Assessing real options in urban surface water flood risk management under climate
- change. Natural Hazards, 94(1), 1–18. (Publisher: Springer Science and Business Media LLC). https://doi.org/10.1007/s11069-018-3349-1 Lübeck, H. (2023). Lübecks Bevölkerung und Wirtschaft wachsen weiter. Retrieved from https://www.luebeck.de/de/presse/pressemeldungen/ view/138971
- MacPherson, L. R., Arns, A., Dangendorf, S., Vafeidis, A. T., & Jensen, J. (2019). A stochastic extreme sea level model for the German Baltic Sea coast. Journal of Geophysical Research: Oceans, 124(3), 2054–2071. https://doi.org/10.1029/2018JC014718
- MacPherson, L. R., Arns, A., Fischer, S., Méndez, F. J., & Jensen, J. (2023). Bayesian extreme value analysis of extreme sea levels along the German Baltic coast using historical information. *Natural Hazards and Earth System Sciences*, 23(12), 3685–3701. https://doi.org/10.5194/ nhess-23-3685-2023
- Mooyaart, L., Jonkman, S., De Vries, P., Van Der Toorn, A., & Van Ledden, M. (2014). Storm surge barrier: Overview and design considerations. *Coastal Engineering Proceedings*, 1(34), 45. https://doi.org/10.9753/icce.v34.structures.45
- NDR, N. R. (2023). Nach Sturmflut an Ostseeküste: Große Schäden, Aufräumarbeiten laufen. Retrieved from https://www.ndr.de/nachrichten/ schleswig-holstein/Nach-Sturmflut-an-Ostseekueste-Grosse-Schaeden-Aufraeumarbeiten-starten,sturmflutostsee106.html
- Nicholls, R. J., Hinkel, J., Lincke, D., & Van Der Pol, T. (2019). Global investment costs for coastal defense through the 21st century. World Bank Policy Research Working Paper.
- Nordhaus, W. (2006). The "stern review" on the economics of climate change (Tech. Rep. No. w12741). National Bureau of Economic Research. https://doi.org/10.3386/w12741
- Oppenheimer, M., Glavovic, B., Hinkel, J., van de Wal, R., Magnan, A. K., Abd-Elgawad, A., et al. (2019). Sea level rise and implications for low lying islands, coasts and communities. In H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. (Eds.), *IPCC Special Report on the Ocean and Cryosphere in a Changing Climate.* (tex.readstatus: read).
- Pachos, K., Huskova, I., Matrosov, E., Erfani, T., & Harou, J. J. (2022). Trade-off informed adaptive and robust real options water resources planning. Advances in Water Resources, 161, 104117. https://doi.org/10.1016/j.advwatres.2021.104117
- Peltier, W. R., Argus, D. F., & Drummond, R. (2015). Space geodesy constrains ice age terminal deglaciation: The global ICE-6G_c (VM5a) model. *Journal of Geophysical Research: Solid Earth*, 120(1), 450–487. (Publisher: American Geophysical Union (AGU)). https://doi.org/10. 1002/2014jb011176
- Powell, W. B. (2011). Approximate dynamic programming. John Wiley & Sons, Inc. https://doi.org/10.1002/9781118029176
- Prahl, B. F., Boettle, M., Costa, L., Kropp, J. P., & Rybski, D. (2018). Damage and protection cost curves for coastal floods within the 600 largest European cities. *Scientific Data*, 5(1), 180034. https://doi.org/10.1038/sdata.2018.34
- Puterman, M. L. (1994). Markov decision processes. Wiley. https://doi.org/10.1002/9780470316887
- Rehdanz, K., Meyerhoff, J., & Wunsch, A. (2022). Meyerhoff, J., Rehdanz, K., & Wunsch, A. Anpassungen an den Klimawandel im Küstenraum: Die Präferenzen der Bevölkerung. Kiel, Rostock: EUCC-D. (OCLC: 1348924848)

Sassone, P. G., & Schaffer, W. A. (1978). Cost-benefit analysis: A handbook (Vol. 182). Academic Press.

- Schwenn, J. (2021). Wie Klimadeiche uns vor dem steigenden Meeresspiegel schützen sollen. NDR (Norddeutscher Rundfunk). Retrieved from https://www.daserste.de/information/wissen-kultur/w-wie-wissen/klimadeiche-100.html
- Skerker, J. B., Zaniolo, M., Willebrand, K., Lickley, M., & Fletcher, S. M. (2023). Quantifying the value of learning for flexible water infrastructure planning. *Water Resources Research*, 59(6), e2022WR034412. https://doi.org/10.1029/2022WR034412
 Stern, N. (2006). Stern review: The economics of climate change.
- Tiggeloven, T., De Moel, H., Winsemius, H. C., Eilander, D., Erkens, G., Gebremedhin, E., et al. (2020). Global-scale benefit–cost analysis of coastal flood adaptation to different flood risk drivers using structural measures. *Natural Hazards and Earth System Sciences*, 20(4), 1025– 1044. https://doi.org/10.5194/nhess-20-1025-2020

- Unterberger, C. (2018). How flood damages to public infrastructure affect municipal budget indicators. *Economics of Disasters and Climate Change*, 2(1), 5–20. https://doi.org/10.1007/s41885-017-0015-0
- van Dantzig, D. (1956). Economic decision problems for flood prevention. *Econometrica: Journal of the Econometric Society*, 24(3), 276. (Publisher: JSTOR). https://doi.org/10.2307/1911632
- Van der Pijl, S., & Oosterlee, C. (2012). An ENO-based method for second-order equations and application to the control of dike levels. *Journal of Scientific Computing*, 50(2), 462–492. (Publisher: Springer. https://doi.org/10.1007/s10915-011-9493-3
- van der Pol, T. D., van Ierland, E., & Weikard, H.-P. (2013). Optimal dike investments under uncertainty and learning about increasing water levels. Journal of Flood Risk Management, 7(4), 308–318. (Publisher: Wiley). https://doi.org/10.1111/jfr3.12063
- Vogelsang, L. G., Weikard, H.-P., Van Loon-Steensma, J. M., & Bednar-Friedl, B. (2023). Assessing the cost-effectiveness of Nature-based Solutions under climate change uncertainty and learning. *Water Resources and Economics*, 43, 100224. https://doi.org/10.1016/j.wre.2023. 100224
- Völz, V., & Hinkel, J. (2023a). Climate learning scenarios for adaptation decision analyses: Review and classification. *Climate Risk Management*, 40, 100512. (Publisher: Elsevier BV). https://doi.org/10.1016/j.crm.2023.100512
- Völz, V., & Hinkel, J. (2023b). Sea level rise learning scenarios for adaptive decision-making based on IPCC AR6. Earth's Future, 11(9), e2023EF003662. https://doi.org/10.1029/2023EF003662
- Völz, V., Hinkel, J., Kupfer, S., MacPherson, L., & Wulff Norrby, C. J. (2024). Learning about sea level rise uncertainty improves coastal adaptation decisions [ComputationalNotebook, Dataset]. https://doi.org/10.5281/zenodo.10262546
- Vousdoukas, M. I., Mentaschi, L., Hinkel, J., Ward, P. J., Mongelli, I., Ciscar, J.-C., & Feyen, L. (2020). Economic motivation for raising coastal flood defenses in Europe. *Nature Communications*, 11(1), 2119. https://doi.org/10.1038/s41467-020-15665-3
- Vousdoukas, M. I., Mentaschi, L., Voukouvalas, E., Verlaan, M., & Feyen, L. (2017). Extreme sea levels on the rise along Europe's coasts. *Earth's Future*, 5(3), 304–323. https://doi.org/10.1002/2016EF000505
- Walker, W. E., Rahman, S., & Cave, J. (2001). Adaptive policies, policy analysis, and policy-making. European Journal of Operational Research, 128(2), 282–289. (Publisher: Elsevier BV). https://doi.org/10.1016/s0377-2217(00)00071-0
- Webster, M., Jakobovits, L., & Norton, J. (2008). Learning about climate change and implications for near-term policy. *Climatic Change*, 89(1–2), 67–85. (Publisher: Springer Science and Business Media LLC). https://doi.org/10.1007/s10584-008-9406-0
- Woodward, M., Kapelan, Z., & Gouldby, B. (2014). Adaptive flood risk management under climate change uncertainty using real options and optimization. *Risk Analysis*, 34(1), 75–92. (Publisher: Wiley tex.readstatus: read). https://doi.org/10.1111/risa.12088
- Wreford, A., Dittrich, R., & Pol, T. D. (2020). The added value of real options analysis for climate change adaptation. *WIREs Climate Change*, 11(3). (Publisher: Wiley tex.readstatus: read). https://doi.org/10.1002/wcc.642
- Zwaneveld, P., Verweij, G., & Van Hoesel, S. (2018). Safe dike heights at minimal costs: An integer programming approach. European Journal of Operational Research, 270(1), 294–301. https://doi.org/10.1016/j.ejor.2018.03.012

References From the Supporting Information

- BGBI. (2010). Verordnung über die Grundsaetze fuer die Ermittlung der Verkehrswerte von Immobilien und der fuer die Wertermittlung erforderlichen Daten. Retreived from www.gesetze-im-internet.de/immowertv_2022/
- DESTATIS, S. B. (2023). Baupreisindizes: Deutschland, Jahre, Messzahlen mit/ohne Umsatzsteuer. Ingenieurbau, Bauarbeiten (Tiefbau). Retrieved from https://www-genesis.destatis.de/genesis//online?operation=table&code=61261-0003&bypass=true&levelindex=0&levelid=1698767963292#abreadcrumb
- Google Earth. (2022). Google Earth. (Publication Title: City of Lübeck)
- Jokar Arsanjani, J., Mooney, P., Zipf, A., & Schauss, A. (2015). Quality assessment of the contributed land use information from OpenStreetMap versus authoritative datasets. In J. Jokar Arsanjani, A. Zipf, P. Mooney, & M. Helbich (Eds.), *OpenStreetMap in GIScience* (pp. 37–58). Springer International Publishing. (Series Title: Lecture Notes in Geoinformation and Cartography). https://doi.org/10.1007/978-3-319-14280-7_3
- Merz, B., Kreibich, H., Schwarze, R., & Thieken, A. (2010). Eview article Assessment of economic flood damage. Natural Hazards and Earth System Sciences, 10(8), 1697–1724. https://doi.org/10.5194/nhess-10-1697-2010
- OpenStreetMap Contributors. (2017). Planet dump. Retrieved from https://planet.osm.org; https://www.openstreetmap.org
- Reese, S., & Markau, H.-J. (2002). Risk handling & natural hazards: New strategies in coastal defense—A case study from Schleswig-Holstein, Germany. In Solutions to coastal disasters '02 (pp. 498–510). American Society of Civil Engineers. https://doi.org/10.1061/40605(258)44
- Schlumberger, J., Ferrarin, C., Jonkman, S. N., Diaz Loaiza, M. A., Antonini, A., & Fatorić, S. (2022). Developing a framework for the assessment of current and future flood risk in Venice, Italy. *Natural Hazards and Earth System Sciences*, 22(7), 2381–2400. https://doi.org/10.5194/nhess-22-2381-2022
- Sieg, T., Vogel, K., Merz, B., & Kreibich, H. (2019). Seamless estimation of hydrometeorological risk across spatial scales. Earth's Future, 7(5), 574–581. https://doi.org/10.1029/2018EF001122
- Statistisches Bundesamt. (2023a). Konjunkturindikatoren. Preisindizes für Bauwerke, Wohngebäude und Nichtwohngebäude. (Publication Title: Preisindizes für Bauwerke: Wohngebäude, Nichtwohngebäude. Tabellen mit Originalwerten und Veränderungsraten.) Retrieved from https:// www.destatis.de/DE/Themen/Wirtschaft/Konjunkturindikatoren/Preise/bpr110.html#241648
- Statistisches Bundesamt. (2023b). Verbraucherpreisindex für Deutschland. Retrieved from https://www.destatis.de/DE/Themen/Wirtschaft/ Konjunkturindikatoren/Basisdaten/vpi001j.html
- Verbraucherschutzzentrale. (2022). Hausratversicherung. (PublicationTitle:AufdenWertderEinrichtungkommt'san) Retrieved from https://www. verbraucherzentrale.de/wissen/geld-versicherungen/weitere-versicherungen/hausratversicherung-auf-den-wert-der-einrichtung-kommts-an-13889